Author's response to reviews

Title: Association Analysis of Chromosome 1 Migraine Candidate Genes

Authors:

Francesca Fernandez (f.fernandez@griffith.edu.au)
Robert P Curtain (robcurtain@hotmail.com)
Micky Ovcaric (m.ovcaric@griffith.edu.au)
John MacMillan (john_macmillan@health.qld.gov.au)
Lyn R Griffiths (l.griffiths@griffith.edu.au)

Version: 2 Date: 1 February 2007

Author's response to reviews: see over
1st February 2007

The Editor
BMC Genomics Journal
Theodora Bloom
Department of Biology
BioMed Central Ltd
Science Navigation Group
Middlesex House
34-42 Cleveland Street
London W1T 4LB
United Kingdom

Dear Madam

Please find enclosed a manuscript entitled “Association Analysis of Chromosome 1 Migraine Candidate Genes” by F. Fernandez, R.P.Curtain R., M. Ovcaric, J. MacMillan and L.R. Griffiths. All contributors have read and approved the submitted version of the manuscript. We respectively request that you consider publishing this manuscript in the “BMC Genomics” as an original research paper.

Correspondence concerning this publication should be directed to:

Professor Lyn Griffiths
Director, Genomics Research Centre
Griffith University, Gold Coast campus
PMB 50 Gold Coast Mail Centre
Queensland  9726  Australia
Fax No: +61-7-5552 9081
Tel No: +61-7-5552 8664
Email: l.griffiths@griffith.edu.au

We thank you in advance for your consideration and await your response.

Yours sincerely

Professor Lyn Griffiths
Director, Genomics Research Centre
Head, School Medical Science
Griffith University

Genomics Research Centre
Genomics Research Centre
School of Medical Science
Gold Coast campus, Griffith University
PMB 50 Gold Coast MC
Queensland 9726
Australia
Telephone:  +61 (0) 7 5552 8808
Facsimile:   +61 (0) 7 5552 9081
Email: l.griffiths@griffith.edu.au
www.genomicsresearchcentre.org